Department of Mathematical Sciences Spring 2002 Carnegie Mellon University

21-256 Multivariate Analysis and Approximation

Assignment 3

Solutions to all the following problems should be written up and handed in to your TA.

Due in recitation on Thursday, February 7, 2002

Stewart:

Section 11.10: Problems 10, 14, 22 Section 12.1: Problems 8, 14, 30 Section 12.5: Problems 2, 12, 22 Walker:

Section 2.2: Problem 4

The address of the web site for this course is

http : //www.andrew.cmu.edu/course/21 − 256/home.html